AGU100 ADVANCING EARTH AND SPACE SCIENCE

Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL084652

Key Points:

- 26 years of surface stress energy input (SSE) are evaluated from observations and a model in the western Arctic Ocean for the first time
- The episodic events of SSE in 2007, 2012 and 2016 triggered step changes of freshwater content in 2007–08, 2012–13 and 2016–17, respectively
- Energy budget analysis reveals that most of the SSE is transformed into potential energy which results in new freshwater state

Supporting Information:

Supporting Information S1

Correspondence to: W. Zhong,

wlzhongouc@ouc.edu.cn

Citation:

Zhong, W., Zhang, J., Steele, M., Zhao, J., & Wang, T. (2019). Episodic extrema of surface stress energy input to the western Arctic Ocean contributed to step changes of freshwater content in the Beaufort Gyre. *Geophysical Research Letters*, 46, 12,173–12,182. https://doi.org/10.1029/2019GL084652

Received 22 JUL 2019 Accepted 18 OCT 2019 Accepted article online 25 OCT 2019 Published online 7 NOV 2019 Corrected 18 DEC 2019

This article was corrected on 18 DEC 2019. See the end of the full text for details.

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Episodic Extrema of Surface Stress Energy Input to the Western Arctic Ocean Contributed to Step Changes of Freshwater Content in the Beaufort Gyre

Wenli Zhong¹, Jinlun Zhang², Michael Steele², Jinping Zhao¹, and Tao Wang³

¹Key Laboratory of Physical Oceanography, Ocean University of China, Qingdao, China, ²Applied Physics Laboratory, University of Washington, Seattle, Washington, USA, ³Laboratory of Marine Environment and Ecology, Ocean University of China, Qingdao, China

Abstract The recent dramatic decline of sea ice in the western Arctic Ocean changes the transfer of momentum across the ice-ocean boundary layer. The surface stress energy input through the surface geostrophic current in the Beaufort Gyre (BG) based on a numerical model is 0.03 mW/m^2 in 1992–2004 versus 0.23 mW/m^2 in 2005–2017. This energy input is primarily concentrated over the southern Canada Basin and the Chukchi Sea. It is 1.38×10^{16} J in observations versus 4.90×10^{16} J in the model in the BG during 2003–2014. We find that some well-known freshwater changes in the BG over 1992–2017 resulted from episodic extrema of energy input in 2007, 2012, and 2016. In particular, most of the energy input in 2007 was transformed into potential energy (57%) which resulted in a new state of freshwater budget. Our study suggests that as of 2016, the BG had not yet reached a saturated freshwater state.

Plain Language Summary The freshwater reservoir of Beaufort Gyre (BG) in the western Arctic Ocean is important both for the local and the global climate. The freshwater content volume of the BG is maintained at a significant high level during the past decade. The reasons for this are not clear. Here we present results of both observation and model to evaluate the surface energy input in the BG for the first time. The corresponding mechanical energy partitions are connected with the freshwater build-up and release processes. Our results revealed that most of the surface energy input in 2007 had transformed into the potential energy, which had triggered the BG into a new normal state with high freshwater content. We highlight the importance of episodic events for the step changes of freshwater content in the BG. The results suggest that as of 2016, the BG had not yet reached a saturated freshwater state. Our results provide a way to predict the future changes of BG freshwater content.

1. Introduction

The Beaufort Gyre (BG) is a wind-driven large-scale ocean circulation in the western Arctic Ocean. Currently, it contains about 23,300 km³ of freshwater in the upper ocean (Proshutinsky & Krishfield, 2019) and is the largest oceanic freshwater reservoir in the northern hemisphere (Carmack et al., 2008). The liquid freshwater budget in the BG showed a dramatic increase in the 2000s (e.g., Giles et al., 2012; Krishfield et al., 2014; Morison et al., 2012; Proshutinsky et al., 2009; Rabe et al., 2014). An interesting phenomenon is that since 2010, the increasing freshwater content in the BG region seemed to be stabilized (e.g., Krishfield et al., 2014; Zhang et al., 2016), while at the same time, the surface forcing shows a weakening trend (Petty et al., 2016). Theoretical studies suggested that lateral eddy flux is the key factor to balance the wind-driven Ekman pumping and thus the stabilization of BG (Manucharyan et al., 2016; Manucharyan & Spall, 2016; Yang et al., 2016), while other studies have considered the important role of geostrophic current in mediating the ice-ocean stress which also contributes to the equilibration of BG (Dewey et al., 2018; Meneghello, Marshall, Timmermans, & Scott, 2018; Zhong et al., 2018). Further, Meneghello, Marshall, Campin, et al., 2018) and Doddridge et al. (2019) noted that the ice-ocean stress feedback is more important than the lateral eddy flux for the equilibration of BG on interannual timescales.

As the BG is a wind-driven system, here we try to address the freshwater issues from a different point of view by investigating its mechanical energy partition. In the ice-free regions, wind energy input through the surface geostrophic current is formulated as: $W = \vec{\tau}_{air-ocean} \cdot \vec{u}_g$ (e.g., Wunsch, 1998; Ferrari & Wunsch, 2009; Zhai et al., 2012; Zhai, 2013), where the air-ocean stress $\vec{\tau}_{air-ocean} = \rho_{air}C_d |\vec{U}_{10} - \vec{u}_{ocean}| (\vec{U}_{10} - \vec{u}_{ocean})$, the



air density $\rho_{air} = 1.25 kg m^{-3}$, the air-ocean drag coefficient $C_d = 0.00125$, and \vec{U}_{10} is the wind speed at 10 m above the sea surface. However, in the BG region where it is partially covered by sea ice, the total surface stress is considered, which is the combination of air-ocean stress and ice-ocean stress scaled by sea ice concentration, that is, $\vec{\tau}_{total} = (1-\alpha)\vec{\tau}_{air-ocean} + \alpha\vec{\tau}_{ice-ocean}$, where $\vec{\tau}_{air-ocean} = \rho_{air}C_d |\vec{U}_{10}|\vec{U}_{10}$ (here the ocean surface velocity of ~5 cm/s is negligible comparing with the wind velocity of ~5 m/s or larger), the ice-ocean stress $\vec{\tau}_{ice-ocean} = \rho_{ocean} C_{iw} \left| \left(\vec{u}_{ice} - \vec{u}_{ocean} \right) \right| R_o \left(\vec{u}_{ice} - \vec{u}_{ocean} \right)$, the ocean density $\rho_{ocean} = 1023 kg m^{-3}$, the ice-ocean drag coefficient $C_{iw} = 0.0055$, and R_o is rotation matrices for ocean (McPhee, 1975, 1980). Thus, it is the surface stress energy (SSE) input through the surface geostrophic current should be considered in the BG as: $W = \vec{\tau}_{total} \cdot \vec{u}_g$, which we denoted as SSE hereafter. Previous studies focusing on the global ocean instead of the Arctic revealed that when the surface current is non-negligible comparing with the wind speed, neglecting the ocean surface current would result in a positive bias in calculations of wind energy input (about 20%-30%) (e.g., Duhaut & Straub, 2006; Hughes & Wilson, 2008; Scott & Xu, 2009). The inclusion of ocean surface current is also important when considering the Ekman dynamics in the BG (Dewey et al., 2018; Meneghello, Marshall, Timmermans, & Scott, 2018; Zhong et al., 2018). However, so far as we know, there are still no studies considering the surface stress energy input in the BG region including the effect of increasing geostrophic current. The basic objective of this study is to investigate the interannual variability of surface stress energy input in the BG region, as well as its role in the interannual variability of freshwater content.

The majority of the wind energy input to surface waves and Ekman layer will be lost in the surface layer and will not enter the deeper ocean (e.g., Huang et al., 2006; von Storch et al., 2007; Zhai et al., 2009). How the surface stress energy input is transferred to the deeper BG interior has yet to be addressed. (By saying deeper BG interior, we refer to the depth that deeper than the Ekman layer.) And its linkage with the accumulation of freshwater in the BG is still unclear. Here, we will address these questions by studying the surface stress energy input and mechanical energy partition in the BG. We used a numerical model and an observational method to quantify the surface stress energy input in the BG. The next section describes the model and observational data that we used. Section 3.1 gives the spatial distribution and temporal variability of SSE. We then evaluate the model's performance in simulating the freshwater content variability of the BG in section 3.2. In section 3.3, the SSE is linked with the mechanical energy partition in the BG and the implication for the freshwater content variability. Our major findings and discussion are presented in section 4.

2. Model and Observations

The monthly mean-derived surface stress $\vec{\tau}_{total}$ and surface geostrophic current \vec{u}_g are used to calculate SSE as: $W = \vec{\tau}_{total} \cdot \vec{u}_g$. Note that although the SSE (low-frequency energy) considered here is only a portion of the total surface stress energy input (including both low- and high-frequency energy) to the ocean, it plays a major role in regulating the large-scale ocean circulation like the BG (e.g., Huang et al., 2006; Zhai et al., 2012). We focus here on monthly mean SSE (this is also the timescale of the observed geostrophic currents) to study the interannual variability of BG, rather than the more high-frequency energy input (e.g., energy input to surface waves, Ekman layer, and near-inertial current) which will mostly dissipate in the surface Ekman layer. We also used the observed geostrophic currents via dynamic ocean topography data (Armitage et al., 2016, 2017) to calculate the observation-based SSE, which is denoted as "OBSm" here. We refer readers to Zhong et al. (2018) for the details of surface stress calculation using observations.

In order to extend the study period to 1992–2017, model results from the Marginal Ice Zone Modeling and Assimilation System (MIZMAS) are used. MIZMAS is a coupled ice-ocean model that assimilates satellite observations of sea ice concentration and sea surface temperature. The ocean model was originally based on the Parallel Ocean Program (Smith et al., 1992) and modified by Zhang and Steele (2007). More detailed information about MIZMAS model components, domain, and grid configuration can be found in Zhang et al. (2016), where extensive validation against in situ observations in the BG was also presented. A caveat is that the model we used here has an average horizontal resolution in the Chukchi and Beaufort Seas of ~10 km and is thus not fully eddy-resolving. However, the model is sufficient to study the large-scale circulations such as the BG (~ 10^3 km) with a timescale longer than months. (This had been validated in Zhang et al.

Geophysical Research Letters



Figure 1. Spatial distribution of the surface stress energy input through the surface geostrophic current estimated from observation and the model. (a) From OBSm using dynamic ocean topography data averaged over 2003–2014, (b) from MIZMAS model averaged over 2003–2014, and from MIZMAS model averaged over 1992–2004 (c) and 2005–2017 (d). Note that the surface geostrophic current in MIZMAS model is derived from the monthly model sea surface height. The pink line encircled region in (a) represents the Beaufort Gyre region. The abbreviations "CS" and "CB" in (a) represent the Chukchi Sea and the Canada Basin, respectively. Gyre; MIZMAS = Marginal ice zone modeling and assimilation system; OBSm = Observation-based SSE.

(2016).) A caveat is that our model is not an eddy-resolving model, so that the eddy kinetic energy (KE) or eddy dissipation might be underestimated. In the model, the surface geostrophic current is derived from the model sea surface height. The model has a vertical resolution of 5 m over the upper 80 m. Both the observed method and model are using the same National Center for Environmental Prediction/National Center for Atmospheric Research reanalysis.

3. Results

3.1. The Surface Stress Energy Input Through the Surface Geostrophic Current in the BG Region

Figures 1a and 1b show that the surface stress energy input through the surface geostrophic current estimated from OBSm and the model are generally similar both in the spatial distribution and magnitude. Significant and strong SSE appears in the northern Chukchi Sea and the southern Canada Basin which indicates strong mechanical energy input into the ocean. The strong SSE in the southern Canada Basin acts like an engine to speed up the flywheel of BG (Proshutinsky et al., 2002), while the strong SSE along the Chukchi Slope and the northern Chukchi Sea might contribute to a stronger Chukchi slope current (Corlett & Pickart, 2017; Li et al., 2019; Spall et al., 2018). In contrast, negative SSE appears in the southern Chukchi Sea and the northern Canada Basin, which indicates areas where the surface stress represents a brake on oceanic KE. How can this happen? There are two possibilities: (i) relatively faster ocean currents rubbing against slower sea ice (Dewey et al., 2018; Doddridge et al., 2019; Meneghello, Marshall, Campin, et al., 2018) or (ii) wind stress forcing that opposes the northward inflow of Pacific Water in the southern Chukchi Sea. The different sign of SSE in the southern versus the northern Canada Basin indicates that mechanical energy goes into the southern limb of BG, while it is lost in the northern limb of BG. The net effect of this inhomogeneous distribution of SSE might contribute to the asymmetric shape of the BG (Regan et al., 2019; Zhong et al.,



Figure 2. Comparison of surface stress (blue line) and surface geostrophic current (red line) in the BG region diagnosed from OBSm (dashed) and MIZMAS model (solid) over the period of 1992–2017. Vgeo is the speed of the surface geostrophic current. (a) and (c) are monthly mean and annual mean, respectively. Comparison of surface stress energy input through surface geostrophic current in the BG region diagnosed from OBSm (pink line) and MIZMAS model (black line) over the period of 1992–2017. (b) and (d) are monthly mean and annual mean, respectively. The calculations are derived from monthly mean variables. The correlation coefficients between each variable pairs are shown as "R" along with the significance level "P" in each panel. MIZMAS = Marginal ice zone modeling and assimilation system; OBSm = Observation-based SSE.

2018). With regard to variability, there is high SSE standard deviation in the Chukchi Sea and the southern Canada Basin (Figure S1), where the surface geostrophic current shows a significant increasing trend (Armitage et al., 2017; Zhong et al., 2018) and is dominated by strong seasonal variability of SSE (Figure S2). The north-south SSE difference in the Canada Basin is larger in the later period (Figures 1c and 1d), relative to the earlier period. Significant increases of SSE also appear in the northern Chukchi Sea.

The MIZMAS model has a high correlation coefficient with the OBSm regarding the surface stress magnitude and surface geostrophic current both in the monthly and annual variability (Figures 2a and 2c). Both the surface stress and surface geostrophic current reveal a strong seasonal variability, with the maximum value usually appearing in fall (Figure S3, also see Meneghello, Marshall, Timmermans, & Scott, 2018; Zhong et al., 2018). The surface geostrophic current shows a dramatic increase of ~2 cm/s since 2007. The model surface geostrophic current is generally a bit larger than the observational results, while the model's surface stress magnitude is generally consistent with observations. No clear increasing trend is revealed in surface stress variance, although there is an exceptional peak in 2007 (Figures 2a and 2c). The SSE shows contrasting variabilities before 2002 and after 2002 (Figures 2b and 2d), with weak fluctuations (generally



Figure 3. Freshwater content volume (10^3 km^3) in the Beaufort Gyre region in 1992–2017 (relative to salinity of 34.8). The light green bar shows July–October model mean, used to compare with the observation (red dots). The dot-line is the monthly mean value from the model. The red error bar is the freshwater content uncertainty due to optimal interpolation from the in situ hydrographic data. The in situ hydrographic data is provided by the Beaufort Gyre Exploration Project that maintained by the Woods Hole Oceanographic Institution.

from -0.5 to 1 mW/m^2) before 2002 and strong fluctuations after (generally from -2 to 2 mW/m^2). Stronger fluctuations indicate the combined effects of faster and retreating sea ice and increasing geostrophic current (e.g., Petty et al., 2016; Zhang et al., 2016). The 2007 SSE peak (with ~9 mW/m² for monthly mean and ~1.5 mW/m² for annual mean) is consistent with the peak of surface stress magnitude and surface geostrophic current (Figures 2a and 2b). The SSE from model is generally larger than that from the OBSm. This may be either due to an overestimation of the surface geostrophic current by the model or due to an underestimation from observations with coarse resolution in space and time. However, the interannual variability of the SSE from the MIZMAS model is similar to that of the observational derived SSE over 1992–2017. This indicates that we can evaluate the interannual variability of SSE from the model.

The most significant increase of SSE appears in fall (November) followed by winter (February) during 2005–2017 compared with 1992–2004 (Figures S2 and S3). This indicates that the retreat of sea ice in the southern Canada Basin during these seasons contributes to a larger surface stress energy input into the ocean. There is also a moderate drop of SSE during spring (May) in the second period, which suggests that the surface stress takes the energy out of the ocean when the ocean drives the ice (Dewey et al., 2018).

3.2. The Step Change of Freshwater Content in the BG

In this section, we evaluate the model's robustness in simulating the freshwater accumulation in the BG. Figure 3 shows that the model-simulated freshwater content has a very high correlation coefficient with the observations on interannual timescales, despite the model's freshwater content bias. Although the model did not capture the exact magnitude of freshwater content volume, it did capture the interannual variability of this wind-driven freshwater build up and release process. Both the model and observations indicate that the freshwater content dramatically increases in 2007 and 2008, followed by a slightly varying plateau of 4 years. The observations show a steep drop of freshwater content in 2013 which then gradually rebounds. In contrast, the model freshwater content seems to be more or less stabilized during 2008–2015 (with a moderate drop in 2013). Both the model and observation show that the freshwater content reaches maxima in 2016 and 2017. Overall, the model is able to represent the interannual variability of freshwater content in the BG, when compared to the observation. This provides us a basic condition to study the mechanical energy partition in the BG system in our following discussions.

The well-known increase of freshwater content in 2007/2008 and the abrupt decrease of freshwater content in 2012/2013 were reported to connect with some extreme events in the BG region. Here we defined the step changes of freshwater content as the change of annual mean freshwater content volume that is larger than 10^3 km³ (which is about half of the standard deviation of freshwater content volume). The first event was the historic 2007 sea ice minimum with a large freshwater availability (Wang et al., 2018), enhanced by strong Ekman convergence (Meneghello, Marshall, Timmermans, & Scott, 2018; Zhong et al., 2018) which led to a sharp increase of freshwater content in 2008. The second event was strong cyclonic winds (storms) that appeared in the Canada Basin in summer of 2012, which limited Ekman convergence in the surface freshwater layer, resulting in a drop of freshwater content in the following year 2013 (Wang et al., 2018). These results are confirmed in Figure 3. In addition, our results reveal another step increase of freshwater content that appears in 2016-2017. These three events are each preconditioned by SSE extrema (Figures S4 and S5), that is, the maxima of SSE in 2007/2008 and 2016/2017 correspond to the step increases of freshwater content in 2008 and 2016, while the minimum of SSE in 2012 corresponds to the step decrease of freshwater content in 2013. There was also a noticeable small bump in 2004, but the intensity of this event is similar to those during 2009-2011 (Figure 2d). The freshwater content increase in 2005 is smaller than our definition for step change, so that the SSE in 2004 does not consider as "an event."

3.3. The Interior Adjustment of Upper Ocean to the Episodic Extrema of Surface Stress Energy Input and Mechanical Energy Partition in the BG

In this section, we will address the mechanical energy partitions in the upper 80 m of the BG. Note that the typical Ekman layer depth in the Arctic Ocean is about 20 m (Cole et al., 2017). We focused on the upper 80 m to reduce the effects of the lateral volumetric injection of Pacific Water (e.g., Shimada et al., 2005; Zhong et al., 2019) since the Pacific Water also provides momentum flux to the BG. Note that the 80 m would eliminate most of the Pacific Winter Water but not the Pacific Summer Water. We employed the basic momentum equation and the rate of change of potential energy Φ to study the surface stress energy input and its mechanical energy partitions in the upper BG (the MIZMAS model results are used for this analysis), that is,

$$\rho\left(\frac{D\,\vec{u}}{Dt} + 2\,\vec{\Omega}\times\vec{u}\right) = -\nabla p - \rho\,\vec{g} + \mu\nabla^2\,\vec{u} \tag{1}$$

$$\frac{\partial(\rho\Phi)}{\partial t} + \nabla \cdot \left(\rho\Phi \ \vec{u}\right) = \rho \ \vec{u} \cdot \nabla\Phi \tag{2}$$

where $\mu \nabla^2 \vec{u} = A_h \nabla_h^2 \vec{u}_h + A_v \nabla_v^2 \vec{u}_h$, A_h and A_v are the horizontal and vertical viscosity, respectively. ∇p is the pressure gradient. After some manipulations by combining the above equations we have:

$$\frac{\partial}{\partial t}(\rho E + \rho \Phi) + R^* = \iint \left(\vec{\tau}_{top} \cdot \vec{u}_{top} - \vec{\tau}_{bot} \cdot \vec{u}_{bot}\right) ds \tag{3}$$

where the KE $\rho E = \rho \ \vec{u} \cdot \vec{u} / 2$, the potential energy (PE) $\rho \Phi = \rho gz$, the area that is used for the integration on the right side of the equation is shown in Figure 1a. Because we are focusing on the upper 80 m, $\vec{\tau}_{bot}$ is the interfacial shear stress (O'Brien & Hurlburt, 1972) between two vertical adjacent model grids at 80 m and 86 m: $\vec{\tau}_{bot} = \rho C_{in} (\vec{u}_{80} - \vec{u}_{86}) (|\vec{u}_{80}| + |\vec{u}_{86}|)/2$, the interfacial drag coefficient $C_{in} = 5 \times 10^{-4}$, \vec{u}_{bot} is the ocean velocity at 80 m. The residual term $R^* = \nabla \cdot (\rho E \ \vec{u} + p \ \vec{u} + \rho \Phi \ \vec{u}) - A_h \nabla^2 E_h + \rho \varepsilon$, where the divergence terms is related with the elevation or depression of ocean surface, $-A_h \nabla^2 E_h$ is the lateral energy input from the Pacific Water injection (or other lateral water injection), and ρz is the dissipation of KE (the eddies' effect). To consider the large-scale BG circulation with a timescale longer than month, $\vec{u}_{top} \cdot \vec{u}_{top} \cdot \vec{u}_{top}$ is used for the following processes: (i) acceleration of the ocean current KE, (ii) enhancement of PE via collection of freshwater which changes the density distribution, (iii) dissipation by the interfacial shear stress $\vec{\tau}_{bot} \cdot \vec{u}_{bot}$, and (iv) dissipation of KE ρz and the divergence terms in the residual term R^* . We do not attempt to evaluate the residual terms in equation (3), because there are many uncertainties in these terms due to the model we used (especially the eddy-related terms). Instead, our study





Figure 4. The mechanical energy budget in 1992–2017. (a) and (b) are the KE derived from DOT data and from model SSH data that averaged over 2003–2014 with an assumed constant potential density of 1025 kg/m³. (c) Area-averaged kinetic energy in the BG from (a) and (b). (d)The integrated surface stress energy input through the surface geostrophic current (SSE) minus the interfacial shear stress work, change of potential energy (Δ PE) and change of kinetic energy (Δ KE) in upper 80 m of the BG region. Because Δ PE is 100–1000 times of the Δ KE, the curve of Δ PE + Δ KE is almost the same as Δ PE. The color lines are the monthly mean, while the color circles are the annual mean. Note that it is an artefact of the annual averaging process that makes the annual mean Δ PE + Δ KE increase from 2006 to 2007 to be about the same as 2007 to 2008.

constitutes a first attempt to evaluate the changes of large-scale surface energy input and its fate in the ocean interior which is linked with the freshwater content variability.

Figures 4a and 4b show that the KE from OBSm and from the model is generally consistent in the spatial distribution, although the model shows large KE and more spatial variability. The variability of the model surface KE is consistent with the observed derived KE, despite that the model values are generally larger (Figure 4c). The KE of BG is increased by $\sim 1 \text{ J/m}^3$ since 2007.

Because we are considering the mechanical energy input and partitions since 1992, the SSE and the interfacial shear stress work W_{int} are integrated over 1992–2017 (Figure 4d). The change of potential energy ΔPE and kinetic energy ΔKE since 1992 are then analyzed, which both shows significant interannual variability (Figure 4d). The ΔPE is rather stable during 1992–1999, gradually increases during 2000–2006, increases steeply during 2007–2008, and then becomes stable again since 2010. The ΔPE and ΔKE also have a strong seasonal variability, with maxima in fall and minima in midyear. The maxima of ΔPE and ΔKE in fall are associated with the maxima of surface stress and surface geostrophic current during this season. Figure 4d shows that the change of potential energy ΔPE in BG is of order 10^2-10^3 times the KE (although both of them increase or decrease together); this is consistent with the findings of Gill et al. (1974) who used a simple two-layer model. There was a large step change of the integrated SSE minus Wint in 2007 (an increase of 3.38 $\times 10^{16}$ J), which corresponds to a dramatic increase of $\Delta PE (1.94 \times 10^{16}$ J) during 2007–2008; thus 57% of SSE is transformed into ΔPE (Figure 4d). This indicates that most of the surface stress energy input to the BG is transformed into PE (of order 10^{16} J), with the interfacial shear stress work W_{int} (of order 10^{12} J) being negligible. The ΔPE has a high correlation coefficient with the freshwater content volume, which indicates that the increase of freshwater content in BG is equivalent to the increase of PE (also see Polyakov et al., 2018). This is reasonable as the accumulation of freshwater steepens the isopycnal slope and thus increases the PE. In this way, the change of potential energy is a reflection of the change of freshwater content volume. The interannual variability of ΔPE could be roughly divided into four periods: 1992–1999 as the stable period, 2000-2006 as the period of gradually increase, 2007-2008 as the period of dramatic increase, and 2009-2017 as the new stable period but with higher ΔPE , which is consistent with periods defined by some previous studies (e.g., Zhang et al., 2016; Zhong et al., 2018).

Because our model is not an eddy-resolving model, the energy partition to the eddy terms might change if we were to use a fully eddy-resolving model. This remains to be investigated. In addition, for equation (3), integrating over a thicker depth range results in a higher percentage of mechanical energy (PE + KE) partition (because the surface energy input penetrates deeper than 80 m). Thus, the percentage of 57% is a lower limit. But note that integrating over larger depth could induce a stronger lateral water injection which we try to avoid.

4. Discussion and Conclusions

For the first time, to our knowledge, we give a robust estimation of the surface stress energy input in the BG from a view of the large-scale gyre circulation. The interannual variability of freshwater budget in the BG depends on low-frequency energy input, which could be quantified by the monthly derived SSE. Before 2003, the SSE is rather small, with larger fluctuations thereafter. There are two significant peaks of SSE in 2007 and 2016 which correspond to step increases of freshwater content in the following years (i.e., 2008 and 2017). During these events, surface stress forcing is generally in the same direction as surface geostrophic currents in the BG, so that the ocean gains mechanical energy (Figure S4). As a first attempt, our results show how much of the surface energy input had transformed into the PE which is equivalent to the freshwater content changes. The extreme of energy input in 2007 is the key contribution for the new state of high freshwater content since 2007 and persisted through 2017. The resulting high level of freshwater content may remain for many years, which Johnson et al. (2018) described as the decadal memory. Thus, the increase and maintenance of BG freshwater content are associated with episodic events of strong SSE. In contrast, the year 2012 is dominated by cyclonic wind and a negative minimum of SSE which results in a step decrease of freshwater in 2013. At this time, the direction of surface stress forcing is generally opposite to that of surface geostrophic current in the BG, so that the ocean loses mechanical energy. These results suggest that the change of freshwater content is a 1 year or less time-lagged response to episodic SSE change (with the understanding that the changes of freshwater content since 2008 are smaller positive or negative freshwater content anomalies (e.g., 2012 and 2016) on the new freshwater state). To some extent, this provides a useful way to predict future change of BG freshwater if we know the SSE of the previous year. The multivear mean of SSE in the BG was 0.03 mW/m^2 in 1992–2004 versus 0.23 mW/m^2 in 2005–2017 (Figure 2). This is a dramatic increase of mechanical energy input in the BG and has thus forced the BG into a "new normal" state with stronger geostrophic current and higher freshwater content.

The total integrated mechanical energy input was 1.38×10^{16} J in observation versus 4.90×10^{16} J in model in the BG during 2003-2014, while the MIZMAS model captures the interannual variability of the mechanical energy input very well (i.e., the change of SSE). MIZMAS results showed that most (57%) of the BG surface stress energy input (SSE) in 2007 is transformed into PE, with the rest dissipated through eddies and contributing to the divergence terms in the residual term of equation (3) which is related with the elevation or depression of ocean surface. In this case, the model can capture the interannual variability of freshwater content very well. Because of the strong baroclinic effects in the BG system (Zhao et al., 2018), the topographic roughness for the energy dissipation of the upper layer is relatively small. On the other hand, some results (Zhao & Timmermans, 2018) suggested that the topographic Rossby waves might be important for the energy dissipation of the BG. This remains to be further investigated. An interesting and important question is "How much freshwater does it take to saturate the BG?" Doddridge et al. (2019) proposed a three-way balance between the wind stress, the ice-ocean governor (the ice-ocean stress feedback), and the eddy fluxes. Ideally, the freshwater content volume should be determined by this three-way balance. The modelintegrated SSE reached a maximum in the early months of 2016, with no further net energy input into the BG after that time (Figure 4d). However, the PE (or freshwater content) continued to increase. This indicates that the peak of SSE in early 2016 is gradually transformed into the PE of BG which is also reflected in the historically high freshwater content in 2017. This suggests that by the year 2016, the BG has not yet reached a state of freshwater content saturation, although 2016 saw a strong SSE. The lagged response of PE to the surface energy input is a result of the inertia of the ocean. The interior ocean would take time to adjust to the changing surface forcing. Our results suggest that with 1-year surface forcing anomaly, the response of the interior ocean would manifest 1 year later. In addition, with an extreme surface energy input like that of 2007, the freshwater content of BG subjects to a decadal memory (Johnson et al., 2018).

For the present study, we investigated on a defined domain of BG and try to evaluate its freshwater state in this domain. However, for the real BG, it could expend its domain that allow more freshwater to be accumulated instead of developing into deeper ocean to reach the saturated state. It is very likely that the BG expands more easily on horizontal direction than to expand to deeper ocean to reach the saturated state. In fact, the observed dynamic ocean topography revealed that the BG is expanding (Regan et al., 2019). This cannot be simulated with those theoretical idealized model simulations (e.g., Doddridge et al., 2019; Manucharyan & Spall, 2016; Yang et al., 2016) which the study region is restricted to a basin domain. Results have showed that the Ekman pumping is relatively stable in recent years (e.g., Meneghello, Marshall, Timmermans, et al., 2018; Zhang et al., 2016; Zhong et al., 2018). And the area of Ekman downwelling is expanding (Figures 7i and 7l in Zhang et al., 2016). While it would take much more energy for the BG to pump the water down (stronger Ekman downwelling), the expanding Ekman downwelling area would favor the BG to converge the surface freshwater from a larger area without a solid boundary. All these lead to an expansion of BG in horizontal direction.

The increases of cyclone activities tend to decrease the surface stress energy input into the BG. That is because the cyclonic wind forcing opposes the anticyclonic BG circulation, thus taking energy out of the ocean and releasing freshwater that resides in the BG. There is an increasing trend of cyclone activity during winter in the Pacific side of the Arctic Ocean (Moore et al., 2018; Zahn et al., 2018), which might contribute to future releases of BG freshwater.

References

- Armitage, T. W., Bacon, S., Ridout, A. L., Thomas, S. F., Aksenov, Y., & Wingham, D. J. (2016). Arctic sea surface height variability and change from satellite radar altimetry and GRACE, 2003–2014. *Journal of Geophysical Research: Oceans*, 121, 4303–4322. https://doi.org/ 10.1002/2015JC011579
- Armitage, T. W. K., Bacon, S., Ridout, A. L., Petty, A. A., Wolbach, S., & Tsamados, M. (2017). Arctic Ocean surface geostrophic circulation 2003–2014. The Cryosphere, 11, 1767–1780. https://doi.org/10.5194/tc-11-1767-2017
- Carmack, E., McLaughlin, F., Yamamoto-Kawai, M., Itoh, M., Shimada, K., Krishfield, R., & Proshutinsky, A. (2008). Freshwater storage in the Northern Ocean and the special role of the Beaufort Gyre. In R. R. Dickson, J. Meincke, & P. Rhines (Eds.), chapter 7 in Arctic-Subarctic Ocean Fluxes: Defining the Role of the Northern Seas in Climate (pp. 145–170). New York: Springer.
- Cole, S. T., Toole, J. M., Lele, R., Timmermans, M. L., Gallaher, S. G., Stanton, T. P., et al. (2017). Ice and ocean velocity in the Arctic marginal ice zone: Ice roughness and momentum transfer. *Elementa: Science of the Anthropocene*, 5, 55. http://doi.org/10.1525/ elementa.241
- Corlett, W., & Pickart, R. S. (2017). The Chukchi slope current. Progress in Oceanography, 153, 50–65. https://doi.org/10.1016/j. pocean.2017.04.005
- Dewey, S., Morison, J., Kwok, R., Dickinson, S., Morison, D., & Andersen, R. (2018). Arctic ice-ocean coupling and gyre equilibration observed with remote sensing. *Geophysical Research Letters*, 45, 1499–1508. https://doi.org/10.1002/2017GL076229
- Doddridge, E. W., Meneghello, G., Marshall, J., Scott, J., & Lique, C. (2019). A Three-way balance in the Beaufort Gyre: The ice-ocean governor, wind stress, and eddy diffusivity. *Journal of Geophysical Research: Oceans*, 124, 3107–3124. https://doi.org/10.1029/ 2018JC014897
- Duhaut, T. H., & Straub, D. N. (2006). Wind stress dependence on ocean surface velocity: Implications for mechanical energy input to ocean circulation. Journal of Physical Oceanography, 36(2), 202–211. https://doi.org/10.1175/JPO2842.1
- Ferrari, R., & Wunsch, C. (2009). Ocean circulation kinetic energy: Reservoirs, sources, and sinks. Annual Review of Fluid Mechanics, 41(1), 253–282. https://doi.org/10.1146/annurev.fluid.40.111406.102139
- Giles, K. A., Laxon, S. W., Ridout, A. L., Wingham, D. J., & Bacon, S. (2012). Western Arctic Ocean freshwater storage increased by winddriven spin-up of the Beaufort Gyre. *Nature Geoscience*, 5(3), 194–197. https://doi.org/10.1038/ngeo1379
- Gill, A. E., Green, J. S. A., & Simmons, A. J. (1974). Energy partition in the large-scale ocean circulation and the production of mid-ocean eddies. *Deep Sea Research*, 21, 499–528.
- Huang, R. X., Wang, W., & Liu, L. L. (2006). Decadal variability of wind-energy input to the world ocean. Deep Sea Research Part II: Topical Studies in Oceanography, 53(1-2), 31–41. https://doi.org/10.1016/j.dsr2.2005.11.001
- Hughes, C. W., & Wilson, C. (2008). Wind work on the geostrophic ocean circulation: An observational study on the effect of small scales in the wind stress. *Journal of Geophysical Research*, 113, C02016. https://doi.org/10.1029/2007JC004371
- Johnson, H. L., Cornish, S. B., Kostov, Y., Beer, E., & Lique, C. (2018). Arctic Ocean freshwater content and its decadal memory of sea-level pressure. *Geophysical Research Letters*, 45, 4991–5001. https://doi.org/10.1029/2017GL076870
- Krishfield, R. A., Proshutinsky, A., Tateyama, K., Williams, W. J., Carmack, E. C., McLaughlin, F. A., & Timmermans, M.-L. (2014). Deterioration of perennial sea ice in the Beaufort Gyre from 2003 to 2012 and its impact on the oceanic freshwater cycle. *Journal of Geophysical Research: Oceans*, 119, 1271–1305. https://doi.org/10.1002/2013JC008999
- Li, M., Pickart, R. S., Spall, M. A., Weingartner, T. J., Lin, P., Moore, G. W. K., & Qi, Y. (2019). Circulation of the Chukchi Sea shelfbreak and slope from moored timeseries. *Progress in Oceanography*, 172, 14–33. https://doi.org/10/1016/j.pocean.2019.01.002
- Manucharyan, G. E., & Spall, M. A. (2016). Wind-driven freshwater buildup and release in the Beaufort Gyre constrained by mesoscale eddies. *Geophysical Research Letters*, 43, 273–282. https://doi.org/10.1002/2015GL065957
 - Manucharyan, G. E., Spall, M. A., & Thompson, A. F. (2016). A theory of the wind-driven Beaufort Gyre variability. Journal of Physical Oceanography, 46(11), 3263–3278. https://doi.org/10.1175/JPO-D-16-0091.1
- McPhee, M. G. (1975). Ice-momentum transfer for the AIDJEX ice model. AIDJEX Bulletin, 29, 93-112.
- McPhee, M. G. (1980). An analysis of pack ice drift in summer. In R. Pritchard (Ed.), Sea ice processes and models (pp. 62–75). Seattle, WA: University of Washington Press.

Acknowledgments

We thank Gianluca Meneghello, Ed Doddridge (MIT) and an anonymous reviewer for their constructive comments and suggestions. We thank Xiaopei Lin (OUC) for valuable discussions and inputs. We also thank Chun Zhou (OUC) for his comments. This study is supported by the National Basic Research Program of China (2015CB953900; 2018YFA0605901), the Kev Project of Chinese Natural Science Foundation (41330960), the National Natural Science Foundation of China (41706211; 41776192), the Office of Naval Research (grant N00014-12-1-0112), the NSF Office of Polar Programs (PLR-1416920, PLR-1503298, PLR-1602985, PLR-1603259, ARC-1203425; NSF-1602926), and NOAA Climate Program Office (NA15OAR4310170, NA15OAR4320063). Wenli Zhong (201606335011) was supported by the China Scholarship Council for his studies in APL. The in situ hydrographic data is provided by the Beaufort Gyre Exploration Project that maintained by the Woods Hole Oceanographic Institution publicly available at Beaufort Gyre Exploration Project website (http://www.whoi.edu/website/beaufortgyre/). The monthly Arctic Dynamic Ocean Topography data are distributed by CPOM (http://www. cpom.ucl.ac.uk/dynamic_topography/). The MIZMAS data are available from the Polar Sciences Center at the University of Washington (https:// pscfiles.apl.uw.edu/zhang/ MIZMAS168x180/).

- Meneghello, G., Marshall, J., Timmermans, M., & Scott, J. (2018). Observations of seasonal upwelling and downwelling in the Beaufort Sea mediated by sea ice. Journal of Physical Oceanography, 48(4), 795–805. https://doi.org/10.1175/JPO-D-17-0188.1
- Meneghello, G., Marshall, J. C., Campin, J.-M., Doddridge, E., & Timmermans, M.-L. (2018). The ice-ocean governor: Ice-ocean stress feedback limits Beaufort Gyre spin-up. *Geophysical Research Letters*, 45, 11–293. https://doi.org/10.1029/2018GL080171
- Moore, G. W. K., Schweiger, A., Zhang, J., & Steele, M. (2018). Collapse of the 2017 winter Beaufort High: A response to thinning sea ice? Geophysical Research Letters, 45, 2860–2869. https://doi.org/10.1002/2017GL076446
- Morison, J., Kwok, R., Peralta-Ferriz, C., Alkire, M., Rigor, I., Andersen, R., & Steele, M. (2012). Changing Arctic Ocean freshwater pathways. *Nature*, 481(7379), 66–70. https://doi.org/10.1038/nature10705
- O'Brien, J. J., & Hurlburt, H. E. (1972). A numerical model of coastal upwelling. Journal of Physical Oceanography, 2, 14–26. https://doi.org/ 10.1175/1520-0485(1972)002<0014:anmocu>2.0.co;2
- Petty, A. A., Hutchings, J. K., Richter-Menge, J. A., & Tschudi, M. A. (2016). Sea ice circulation around the Beaufort Gyre: The changing role of wind forcing and the sea ice state. *Journal of Geophysical Research: Oceans*, 121, 3278–3296. https://doi.org/10.1002/ 2015JC010903
- Polyakov, I. V., Pnyushkov, A. V., & Carmack, E. C. (2018). Stability of the Arctic halocline: a new indicator of Arctic climate change. Environmental Research Letters, 13(12), 125008. https://doi.org/10.1088/1748-9326/aaec1e

Proshutinsky, A., Bourke, R. H., & McLaughlin, F. A. (2002). The role of the Beaufort Gyre in Arctic climate variability: Seasonal to decadal climate scales. *Geophysical Research Letters*, 29(23), 2100. https://doi.org/10.1029/2002GL015847

Proshutinsky, A., & Krishfield, R. (2019). In a spin: New insights into the Beaufort Gyre. *Eos*, *100*. https://doi.org/10.1029/2019EO119765 Proshutinsky, A., Krishfield, R., Timmermans, M. L., Toole, J., Carmack, E., McLaughlin, F., et al. (2009). Beaufort Gyre freshwater

- reservoir: State and variability from observation. Journal of Geophysical Research, 114, C00A10. https://doi.org/10.1029/2008JC005104 Rabe, B., Karcher, M., Kauker, F., Schauer, U., Toole, J. M., Krishfield, R. A., et al. (2014). Arctic Ocean basin liquid freshwater storage trend 1992–2012. Geophysical Research Letters, 41, 961–968. https://doi.org/10.1002/2013GL058121
- Regan, H. C., Lique, C., & Armitage, T. W. K. (2019). The Beaufort Gyre extent, shape, and location between 2003 and 2014 from satellite observations. Journal of Geophysical Research: Oceans, 124, 844–862. https://doi.org/10.1029/2018JC014379
- Scott, R. B., & Xu, Y. (2009). An update on the wind power input to the surface geostrophic flow of the world ocean. *Deep Sea Research*, 56(3), 295–304. https://doi.org/10.1016/j.dsr.2008.09.010
- Shimada, K., Itoh, M., Nishino, S., McLaughlin, F., Carmack, E., & Proshutinsky, A. (2005). Halocline structure in the Canada Basin of the Arctic Ocean. *Geophysical Research Letters*, 32, L03605. https://doi.org/10.1029/2004GL021358

Smith, R. D., Dukowicz, J. K., & Malone, R. C. (1992). Parallel ocean general circulation Modeling. *Physica D*, 60(1-4), 38–61. https://doi. org/10.1016/0167-2789(92)90225-C

Spall, M. A., Pickart, R. S., Li, M., Itoh, M., Lin, P., Kikuchi, T., & Qi, Y. (2018). Transport of Pacific Water into the Canada Basin. Journal of Geophysical Research: Oceans, 123, 7453–7471. https://doi.org/10.1029/2018JC013825

von Storch, J. S., Sasaki, H., & Marotzke, J. (2007). Wind-generated power input to the deep ocean: An estimate using a 1/10 general circulation model. *Journal of Physical Oceanography*, *37*(3), 657–672. https://doi.org/10.1175/JPO3001.1

Wang, Q., Wekerle, C., Danilov, S., Koldunov, N., Sidorenko, D., Sein, D., et al. (2018). Arctic sea ice decline significantly contributed to the unprecedented liquid freshwater accumulation in the Beaufort Gyre of the Arctic Ocean. *Geophysical Research Letters*, 45, 4956–4964. https://doi.org/10.1029/2018GL077901

Wunsch, C. (1998). The work done by the wind on the oceanic general circulation. Journal of Physical Oceanography, 28(11), 2332–2340. https://doi.org/10.1175/1520-0485(1998)028<2332:TWDBTW>2.0.CO;2

- Yang, J., Proshutinsky, A., & Lin, X. (2016). Dynamics of an idealized Beaufort Gyre: 1. The effect of a small beta and lack of western boundaries. Journal of Geophysical Research: Oceans, 121, 1249–1261. https://doi.org/10.1002/2015JC011296
- Zahn, M., Akperov, M., Rinke, A., Feser, F., & Mokhov, I. I. (2018). Trends of cyclone characteristics in the Arctic and their patterns from different reanalysis data. *Journal of Geophysical Research: Atmospheres*, 123, 2737–2751. https://doi.org/10.1002/2017JD027439
- Zhai, X. (2013). On the wind mechanical forcing of the ocean general circulation. Journal of Geophysical Research: Oceans, 118, 6561–6577. https://doi.org/10.1002/2013JC009086
- Zhai, X., Greatbatch, R. J., Eden, C., & Hibiya, T. (2009). On the loss of wind induced near-inertial energy to turbulent mixing in the upper ocean. Journal of Physical Oceanography, 39(11), 3040–3045. https://doi.org/10.1175/2009JPO4259.1
- Zhai, X., Johnson, H. L., Marshall, D. P., & Wunsch, C. (2012). On the wind power input to the ocean general circulation. Journal of Physical Oceanography, 42, 1357–1365. https://doi.org/10.1175/JPO-D-12-09.1

Zhang, J., & Steele, M. (2007). The effect of vertical mixing on the Atlantic water layer circulation in the Arctic Ocean. *Journal of Geophysical Research*, *112*, C04S04. https://doi.org/10.1029/2006JC003732

Zhang, J., Steele, M., Runciman, K., Dewey, S., Morison, J., Lee, C., et al. (2016). The Beaufort Gyre intensification and stabilization: A model-observation synthesis. *Journal of Geophysical Research: Oceans*, *121*, 7933–7952. https://doi.org/10.1002/2016JC012196

Zhao, B., & Timmermans, M.-L. (2018). Topographic Rossby waves in the Arctic Ocean's Beaufort Gyre. *Journal of Geophysical Research: Oceans*, 123, 6521–6530. https://doi.org/10.1029/2018JC014233

Zhao, M., Timmermans, M.-L., Krishfield, R., & Manucharyan, G. (2018). Partitioning of kinetic energy in the Arctic Ocean's Beaufort Gyre. *Journal of Geophysical Research: Oceans*, *123*, 4806–4819. https://doi.org/10.1029/2018JC014037

Zhong, W., Steele, M., Zhang, J., & Cole, S. T. (2019). Circulation of Pacific Winter Water in the western Arctic Ocean. Journal of Geophysical Research: Oceans, 124, 863–881. https://doi.org/10.1029/2018JC014604

Zhong, W., Steele, M., Zhang, J., & Zhao, J. (2018). Greater role of geostrophic currents in Ekman dynamics in the western Arctic Ocean as a mechanism for Beaufort Gyre Stabilization. *Journal of Geophysical Research: Oceans, 123*, 149–165. https://doi.org/10.1002/ 2017JC013282

Erratum

In the originally published version of this article, a sentence was incorrectly set at the end of the Abstract, the Plain Language Summary was omitted, and two subscripts were not set correctly in the in-line math under equation (2). These errors have since been corrected, and the present version may be considered the authoritative version of record.